Author's response to reviews

Title: National, regional, and global trends in adult overweight and obesity

Authors:

Gretchen A Stevens (stevensg@who.int)
Gitanjali M Singh (gsingh@hsph.harvard.edu)
Yuan Lu (yuanlu@hsph.harvard.edu)
Goodarz Danaei (gdanaei@hsph.harvard.edu)
John K Lin (john.lin@post.harvard.edu)
Mariel M Finucane (mfinucan@hsph.harvard.edu)
Adil N Bahalim (adilnb@gmail.com)
Russell K McIntire (rkmcinti@indiana.edu)
Hialy R Gutierrez (hialy.gutierrez@gmail.com)
Melanie Cowan (cowanm@who.int)
Christopher J Paciorek (paciorek@stat.berkeley.edu)
Leanne Riley (rileyl@who.int)
Majid Ezzati (majid.ezzati@imperial.ac.uk)

Version: 3 Date: 15 September 2012

Author's response to reviews:

Dear Kenji,

Thank you for the conditional acceptance of our paper, now titled “National, regional, and global trends in adult overweight and obesity”, and the helpful suggestions from the Editors. We have implemented the suggestions in the revised version.

We look forward to the decision on the manuscript and would be happy to provide additional information you may need.

Sincerely,

Majid Ezzati